Craig Chambers CSE 401 Introduction to Compiler Construction

Readings and Assignment #9

By: Read:

M 5/22 Reminder: Sections 9.4, 10.1, 10.2

W 5/24 Section 10.4, 10.5

F 5/26 Section 10.6

W 5/31 Sections 9.5, 9.7

By: Do:

F 5/26 Exercises 9.1e, 9.2e, 9.3ce, 9.5bce (you may assume the function calls use the PL/0 calling convention).

F 6/2 Complete the extended PL/0 compiler by implementing the code generation routines for your new AST classes. As usual, turn in printouts of the changed code, with changes highlighted. Turn in commented printouts of your test programs, along with assembly code output and logs from running the generated program demonstrating that your compiler generates correct code; run the extended compiler with the “-a” option to print the assembly code.

As usual, you will be graded on correctness of your implementation, on clarity and good design of your implementation, and on sufficiency of your test cases. In particular, be sure that you test the differences between call-by-reference and call-by-value, and that references to up-level variables work, whether the up-level variable was declared as a local, a call-by-value formal, or a call-by-reference formal.

Note:

M 5/29 MEMORIAL DAY - No class